food & nutrition (

REVIEW ARTICLE

Legume consumption in adults and risk of cardiovascular disease and type 2 diabetes: a systematic review and meta-analysis

Birna Thorisdottir¹*[®], Erik Kristoffer Arnesen²[®], Linnea Bärebring³[®], Jutta Dierkes^{4,5,6}[®], Christel Lamberg-Allardt⁷[®], Alfons Ramel¹[®], Bright I. Nwaru⁸[®], Fredrik Söderlund⁹[®] and Agneta Åkesson⁹[®]

¹Faculty of Food Science and Nutrition, University of Iceland, Reykjavik, Iceland; ²Department of Nutrition, Institute of Basic Medical Sciences, University of Oslo, Oslo, Norway; ³Department of Internal Medicine and Clinical Nutrition, Institute of Medicine, Sahlgrenska Academy, University of Gothenburg, Gothenburg, Sweden; ⁴Centre for Nutrition, Department of Clinical Medicine, University of Bergen, Bergen, Norway; ⁵Mohn Nutrition Research Laboratory, Department of Clinical Science, University of Bergen, Bergen, Norway; ⁷Department of Medical Biochemistry and Pharmacology, Haukeland University Hospital, Bergen, Norway; ⁷Department of Food and Nutrition, University of Helsinki, Helsinki, Finland; ⁸Krefting Research Centre, Institute of Medicine, University of Gothenburg, Gothenburg, Sweden; ⁹Unit of Cardiovascular and Nutritional Epidemiology, Institute of Environmental Medicine, Karolinska Institutet, Stockholm, Sweden

Popular scientific summary

- Evidence from observational studies was suggestive of null associations between legume consumption and CVD and T2D.
- Meta-analyses of RCTs suggested a protective effect of legume consumption on established risk factors for CVD and T2D.
- Legume consumption may be recommended as a substitute for meat in sustainable diets and as part of a diverse and healthy dietary pattern for CVD and T2D prevention.

Abstract

Objectives: This study aimed to systematically review the evidence for associations between consumption of legumes and cardiovascular disease (CVD), type 2 diabetes (T2D) and their risk factors among healthy adults. **Methods:** We searched MEDLINE, Embase, Cochrane Central Register of Controlled Trials, and Scopus up to 16 May 2022 for \geq 4 weeks long randomized (RCT) and non-randomized controlled trials and prospective cohort studies with follow-up \geq 12 months, assessing legume intake (beans/lentils/peas/soybeans, excluding peanuts and legume-products/protein/powder/flour) as the intervention or exposure. Outcomes were CVD, coronary heart disease (CHD), stroke, T2D and in intervention trials only: changes in blood lipids, glycemic markers, and blood pressure. Risk of bias (RoB) was evaluated with Cochrane's RoB2, ROBINS-I, and US Department of Agriculture (USDA)'s RoB-NObS. Effect sizes were pooled using random-effects meta-analyses and expressed as relative risk or weighed mean differences with 95% confidence intervals, heterogeneity quantified as *P*. The evidence was appraised according to World Cancer Research Fund's criteria.

Results: Of the 181 full-text articles assessed for eligibility, 47 were included: 31 cohort studies (2,081,432 participants with generally low legume consumption), 14 crossover RCTs (448 participants), one parallel RCT and one non-randomized trial. Meta-analyses of cohort studies were suggestive of null associations for CVD, CHD, stroke and T2D. Meta-analyses of RCTs suggested a protective effect on total cholesterol (mean difference -0.22 mmol/L), low density lipoprotein (LDL)-cholesterol (-0.19 mmol/L), fasting glucose (-0.19 mmol/L), and HOMA-IR (-0.30). Heterogeneity was high ($I^2 = 52\%$ for LDL-cholesterol, >75% for others). The overall evidence for associations between consumption of legumes and risk of CVD and T2D was considered *limited – no conclusion*.

Conclusion: Legume consumption was not found to influence risk of CVD and T2D in healthy adult populations with generally low legume consumption. However, protective effects on risk factors, seen in RCTs, lend some support for recommending legume consumption as part of diverse and healthy dietary patterns for prevention of CVD and T2D.

Keywords: fabaceae; pulses; beans; lentils; Peas; human nutrition; cardiovascular diseases; diabetes mellitus type 2; non-communicable diseases

To access the supplementary material, please visit the article landing page

Received: 29 September 2022; Revised: 17 April 2023; Accepted: 20 April 2023; Published: 30 May 2023

The global burden of disease study 2017 considered low legume intake a preventable risk factor for non-communicable diseases (1). The term 'legumes' includes various types of beans, lentils, peas, and soybeans. In addition, peanuts classify botanically as legumes, however in nutrition science they are classified as nuts. Legumes are excellent sources of essential amino acids, complex carbohydrates and fiber, are generally low in fat and saturated fatty acids, rich in micronutrients such as potassium, magnesium, folate, iron and zinc, as well as many bioactive compounds such as phytochemicals (2–5). From a sustainability perspective, legumes are considered important sources of dietary protein, as they have low greenhouse gas and water footprints, enrich the soil through nitrogen fixation, and reduce the need for fertilizers (6–8).

While legumes are used for human consumption worldwide, recent data indicate a universal lack of enough legumes in diets in most parts of the world, excluding some parts of Latin America, south Asia, and sub-Saharan Africa (1, 9). In Finland and Sweden, mean adult legume consumption is approximately 12 g/day and may be even lower in other Nordic and Baltic countries (10). The increasingly recognized need for the global population to shift towards a more sustainable, plant-based diet has driven an expansion in available research on health effects of legumes over the past years. This, along with the absence of specific recommendations on legume consumption in the 2012 edition of the Nordic Nutrition Recommendations (NNR); an emphasis in the NNR update for 2022 on healthy food groups that may act as substitutes for meat in sustainable diets; and an interest in possible associations between diets and cardiovascular disease (CVD) and type 2 diabetes (T2D) and associated biomarkers such as blood lipids, glycemic markers and blood pressure, all contributed to this being a prioritized topic for a systematic review in updating the NNR for 2022 (11).

Therefore, we performed a systematic review and meta-analysis of intervention trials and observational studies investigating the role of consumption of legumes in the development of CVD and T2D as well as their risk factors among generally healthy adults.

Methods

This systematic review followed the guidelines developed for the 2022 revision of the NNR (12, 13) and preferred reporting for systematic reviews (14, 15). A study protocol was published prior to article selection in the PROSPERO database (https://www.crd.york.ac.uk, CRD42021250084). This systematic review is part of the NNR 2022 project, funded by the Nordic Council of Ministers and governmental food and health authorities of Norway, Finland, Sweden, Denmark, and Iceland (16).

Eligibility criteria

The research question was developed by the NNR 2022 Committee and the NNR Systematic Review Centre (i.e. the authors) (Supplementary Table 1). The population of interest were adults in settings relevant for the general population in the Nordic and Baltic countries (i.e. free-living and generally healthy). Therefore, studies on patient groups (e.g. participants with established CVD or T2D at baseline) and weight-loss trials were excluded. The intervention/exposure of interest was legumes, excluding peanuts, and the comparator of interest were different intake levels in observational studies, and usual diet or other comparator in intervention studies. Legumes could be any type of fresh or cooked beans, lentils, peas, or soybeans consumed as a food. Specific legume types should be explored, if possible. Studies exploring processed legume products, such as legume protein/powder, legume flour, soy products (e.g. tofu), legume oils and legume snacks (with the exception of soynuts made from mature soybeans in intervention studies), were excluded, as were studies where the intervention/exposure of interest could not be separated from peanuts or vegetables. Substitution analyses (e.g. of replacing meat with legumes) were not included in this review. The outcomes of interest were incidence (non-fatal or fatal) of, or mortality from, CVD, coronary heart disease (CHD) and stroke, T2D incidence, and in intervention studies: changes in blood lipids, blood pressure, fasting glucose or insulin, and insulin resistance (homeostatic model assessment, HOMA-IR). Studies were included if they provided multivariable-adjusted risk estimates [risk ratios (RR), hazard ratios (HR), or odds ratios (OR)] or mean differences (intervention studies) with corresponding data to calculate the variance. Eligible randomized (RCT) and non-randomized controlled trials should have at least 4 weeks duration and prospective cohort studies should have at least 12 months of follow-up.

Search strategy

A comprehensive literature search of MEDLINE, Embase, Cochrane Central Register of Controlled Trials and Scopus was performed at the Library of Medicine and Science at the University of Oslo, Norway up to the search date, initially on 2 May 2021, updated on 16 May 2022. The search strategy (Supplementary file 2) was developed in collaboration with the authors (led by BT and EKA) and peer-reviewed by research librarians at the Karolinska Institute in Stockholm, Sweden. Reference lists of relevant retrieved articles were also screened to identify additional articles. There were no publication date or language limitations in the search. Grey literature and unpublished study searches were not performed.

Article selection and data collection

Pairs of two authors (CL-A, LB initial search, BT, AR updated search) were independently screened and selected studies for inclusion. The screening of titles/abstracts was performed in the web tool Rayyan (https://rayyan.qcri. org) in a blinded fashion. Discrepancies were resolved by discussion or by a third author (AÅ) both after title/ abstract screening and full-text screening.

Data from full-text articles was extracted into pre-specified Excel extraction forms by pairs of two authors working independently (BIN, JD, FS initial search, BT, AR updated search). Among the data extracted were study design, information on participants and recruitment, exposure/interventions and controls, assessment of outcomes, follow-up, confounders, and results. When duplicate publications from the same study were identified, we included the report that included the largest number of cases for each endpoint of interest. In a few instances, study authors were contacted in an attemp to retrieve data considered necessary for meta-analyses.

RoB assessment

Risk of bias of each included study was independently assessed as low, moderate/some concerns, serious or critical by two authors (EAR initial search, BT, AR updated search) using tools relevant for the different study types. Cochrane's Risk of Bias 2 (RoB2) (17) was used for individually randomized, parallel-group trials, considering the RoB in five domains: arising from the randomization process, due to deviations from the intended interventions and missing outcome data, in the measurement of the outcome, and in selection of the reported result. A supplement to the main RoB2 (18) was used for crossover trials, additionally considering the RoB from period and carryover effects. Cochrane's Risk of Bias in Nonrandomized studies - of interventions (ROBINS-I) (19) was used for non-randomized intervention studies, and the 'Risk of Bias for Nutrition Observational Studies' (RoB-NObS) (20) developed by the USDA Nutrition

Evidence Systematic Review (NESR) team was used for observational studies. Both ROBINS-I and RoB-NObS consider the RoB in seven domains, although with slightly different wording in signaling questions: due to confounding, in selection of participants, in classification of interventions/exposures, due to deviations from intended interventions/exposures, due to missing data, in measurement of outcomes, and in selection of the reported result. RoB was visualized by using the web app Risk-of-bias VISualization (robvis) (21).

Synthesis methods

The evidence was synthesized qualitatively, based on study characteristics, context, strengths and limitations, heterogeneity, and relevance. Meta-analyses were performed if deemed appropriate to combine/pool the different studies, but only when more than three independent RCTs or five cohort studies with sufficient homogenous data existed (12). These conditions were met for total CVD and CVD mortality, total CHD and CHD incidence, total stroke and stoke incidence, T2D incidence, total cholesterol (TC), low density lipoprotein cholesterol (LDL-C), high density lipoprotein cholesterol (HDL-C), triglycerides (TG), fasting glucose, insulin, and HOMA-IR. All meta-analyses were performed using a random-effects model and the meta-analysis approach followed recommendations from the Agency for Healthcare Research and Quality (AHRQ) and the Cochrane's Handbook (22, 23). In studies only reporting results stratified by sex or menopausal status, these estimates were first pooled within each study by an inverse variance fixed-effect model. Stratified analyses were performed if data allowed, by separating by incidence versus mortality in cohort studies, and further by legume type in both observational studies and RCTs. Sensitivity analyses were explored by overall RoB (RCTs alone).

For observational studies, the fully adjusted risk estimates [HR/RR; OR was converted into RR (24)] and 95% confidence intervals were log-transformed and summarized to assess the highest versus lowest consumption categories and dose-response relationships. For the dose-response analysis, additional required inputs for each category of exposure were the median exposure amount, person-years of follow-up, and number of cases. For estimating median exposure amounts in g/day in studies with an alternative reporting (such as servings/ day or times/week), methods from Greenland 1987 were used (25). Consistent with most regional serving sizes (26) and in line with previous meta-analyses (27-29), we assumed that one serving of legumes equaled 100 g, if not otherwise specified. Dose-response effects/relationships were assessed by a one-stage approach estimating a pooled regression slope and standard error, from the dose-specific risk estimates within the studies included in the meta-analysis, thus indicating the association between

a unit increase in legume intake and the respective outcomes with the lowest consumption category (usually 0 g/day) as reference (30). Potentially non-linear dose-response associations were explored by restricted cubic splines with knots fixed at the 10th, 50th and 90th percentiles of the exposure (31).

For RCTs, mean differences and their standard deviations (SDs) between the intervention and control group at the end of the intervention period were the primary effects of interest. For RCTs with more than one legume intervention arm, the results were first pooled into one intervention group with a fixed-effect meta-analysis, to avoid double-counting of participants (32). For crossover RCTs, we used results from paired analysis accounting for intra-individual correlation as reported, or calculated SD with a correlation coefficient of 0.6, which is considered a conservative estimate (33). Units were converted if needed: TC, LDL-C and HDL-C from mg/dL to mmol/L by dividing by 38.67, TG from mg/dL to mmol/L by dividing by 88.57, glucose from mg/dL to mmol/L by dividing by 18.02 and insulin from µIU/mL to pmol/L by multiplying by 6.00.

For all meta-analyses, potential heterogeneity between studies was assessed using the I^2 statistic. To assess small study effects, visual inspection of funnel plots and Egger's regression test (significance level P > 0.1) were evaluated if there were at least 10 effect estimates (34). Meta-analyses were performed with the 'meta' command in Stata v17 (StataCorp LLC, College Station, Texas, USA).

Strength of evidence assessment

Strength of evidence was appraised based on RoB, quantity, consistency, and precision in the evidence, according to the World Cancer Research Fund's grading:

'Convincing', 'Probable', 'Limited - suggestive, 'Limited - no conclusion', 'Substantial effects unlikely' (16). According to this classification, several conditions must be met for the body of evidence to be judged as convincing, that is, strong enough to support a causal relationship or substantial effects unlikely, that is strong enough to support that there is a convincing absence of a causal relationship. The evidence is considered as probable when it is strong enough to support that there is a probable causal relationship, with evidence from at least two independent cohort studies, no unexplained heterogeneity between- or within-study types, good-quality studies to confidentially exclude the possible random or systematic errors, and evidence for biological plausibility. The evidence is considered *limited - suggestive* when there is evidence suggestive of a consistent direction of effect, and evidence for biological plausibility. If the evidence is so limited that no firm conclusion can be made or there is inconsistency of direction of effect it is considered limited - no conclusion.

Results

The database searches yielded 10,771 articles after deduplication, of which 10,599 were excluded after title/ abstract screening. Nine additional records were identified by screening reference lists of retrieved articles and other systematic reviews. Of the 181 full-text articles evaluated, 47 met the criteria to be included in the review and 42 were included in quantitative synthesis (meta-analysis). Figure 1 gives the flowchart for the literature screening and Supplementary Table 2 lists the 134 articles excluded after full text screening.

Overview of included studies

In total, 31 articles reporting possible observational associations between legume intake and cardiovascular endpoints (n = 21) or T2D (n = 10) and 16 articles reporting results from legume interventions on cardiometabolic risk factors were included (Table 1). They report results from 25 unique prospective cohorts, prospective observational analysis of participants from one RCT, as well as 13 unique crossover RCTs, one parallel RCT and one non-randomized trial.

Characteristics of observational studies

Summarizing from Table 1 and Supplementary Table 3, the observational studies analyzed 2,081,432 participants (69% women) from Europe, USA, Australia, China, Iran, Japan, and South Korea, and one study combined data from 18 countries in seven geographical regions. Except for the PREDIMED cohort, the participants were obtained from generally healthy populations. One US cohort included a large proportion of vegetarian participants, but results were only presented for the overall population (38). The mean age was 53 years at baseline, ranging in individual studies from 38 to 75 years. Follow-up time ranged from 4 to 28 years in individual studies (mean 12 years) and person-years ranged from 6,986 to 6,170,299 in the studies for which they were reported. Most observational studies (n = 20) were publicly funded.

Diets were assessed by food frequency questionnaire in all but the Nordic studies, which used a diet history (47, 58). More than half of the studies (n = 17) assessed diet only at baseline, while others had two (n = 6) or more (n = 8) diet assessments. The exposure assessed was most often total legumes (beans, lentils, peas, occasionally soybeans), while studies from Japan, South Korea and one study from China only assessed intake of soybeans (43, 44, 48, 54, 64, 65). By a crude approximation we estimated the median intake in the studies to be 20 g/day, but it varied between studies and geographical regions. The Finnish study and the European EPIC study reported medians below 10 g/day, while the PURE study including 18 countries worldwide and



Fig. 1. Flowchart of study selection.

From: Page MJ, McKenzie JE, Bossuyt PM, Boutron I, Hoffmann TC, Mulrow CD, et al. The PRISMA 2020 statement: an updated guideline for reporting systematic reviews. BMJ 2021;372:n71. doi: 10.1136/bmj.n71. For more information, visit: http://www.prisma-statement.org/.

studies from South Korea and Australia reported medians of 40–50 g/day (43, 46, 47, 51, 53, 59).

Outcomes assessed were CVD (5 studies reporting incidence, 7 studies reporting mortality), CHD (8 studies reporting incidence, 3 studies reporting mortality), stroke (6 studies reporting incidence, 3 studies reporting mortality), and T2D (10 studies). In total, 51,454 cardiovascular events and 15,992 diabetes events occurred. The CVD endpoints were most often assessed by reviewing medical records and/or death certificates. In some studies, especially those on T2D, participants or their families self-reported the outcome. Most often measures were taken to check the reported outcomes using official records or other sources. Most studies adjusted for important confounders including age, sex, body mass index, smoking, alcohol use, physical activity, energy intake, relevant medication use or history/family history of relevant diseases or conditions.

Characteristics of intervention studies

Summarizing from Table 1 and Supplementary Table 4, the crossover RCTs analyzed 448 participants (58% women) from Australia, New Zealand, USA, Canada, Iran, and the Netherlands, the parallel-group RCT analyzed 25 participants from Iran and the non-randomized trial analyzed 242 participants from China, although the number reporting was unclear in that article. Most trials included generally healthy participants, some with hypercholesterolemia or metabolic syndrome (MetS), but did not include participants using lipid-lowering medications. The age range was wide (≤ 30 to ≥ 65 years) in 10 studies, one studied young athletes, and the remaining five studied middle aged or older adults. The duration of the crossover RCTs ranged from 4 to 16 weeks (mean 7 weeks), while the parallel-group and non-randomized trials had a duration of 12 weeks. Blood lipids were assessed in 14 studies, glucose or insulin or HOMA-IR in 11 studies and blood pressure in 2 studies. Most intervention studies (n = 8) were industry funded.

The interventions were consumption of either a mixture of legumes (n = 4) or individual types; beans (n = 5), soybeans (n = 3), chickpeas (n = 3) and individual effects of pinto beans and black-eyed peas (n = 1). Various methods were used to assess adherence to the intervention. Exposure amounts were in the ranges of 37–156 g/day (dry weight) for mixed legumes, 50–377 g/day (cooked weight) for beans, 30–67 g/day for soynuts and 130–140 g/day (cooked weight) for chickpeas, pinto beans, and black-eyed peas. A few of

Table 1. Charact	eristics of included	studies.								
Article, reference	Study design (cohort name, if applicable)	Country	Subjects	Age at entry	Number analyzed (% women)	Exposure	Follow-up (y) / Duration (weeks)	Outcomes	Funding	Risk of bias
Observational stu	Idies, cardiovascular	* outcomes								
Bernstein 2012 (89)	PC (NHS, HPFS)	USA	Female registered nurses; male health care professionals	30-55 (NSH), 40-75 (HPFS)	127160 (66)	Legumes	26 (NHS), 22 (HPFS)	Stroke	Public	Moderate
Blekkenhorst 2017 (90)	PC (PLSAW)	Australia	Elderly women	≥70	1226 (100)	Legumes	15	CVD	Public	Moderate
Farvid 2017 (91)	PC (GCS)	Iran	General population	36-85	42403 (57)	Legumes	=	CVD, CHD, Stroke	Mixed	Moderate
Fraser 1992 (92)	PC (AHS)	NSA	Seventh-day Adventists	≥25	26743 (63)	Legumes	6	CHD	Public	Serious
Fung 2018 (93)	PC (NHS, NHS II, HPFS)	USA	Female registered nurses; male health care professionals	30-55 (NHS), 25-42 (NSH II), 40-75 (HPFS)	212142 (79)	Legumes	28 (NHS), 20 (NHS II), 24 (HPFS)	CHD	Public	Moderate
Golzarand 2022 (51)	PC (TLGS)	Iran	General population	20-69	2863 (56)	Legumes	=	CVD	Not reported	Moderate
Haring 2014 (94)	PC (ARIC)	NSA	General population	45-65	12066 (56)	Legumes	22	CHD	Public	Moderate
Haring 2015 (95)	PC (ARIC)	USA	General population	45-65	11601 (56)	Legumes	22	Stroke	Public	Moderate
Im & Park 2021 (41)	PC (Ansan-Ansung)	South Korea	General population, urban and rural	40-69	4713 (100)	Soybeans	7	CVD	Public	Moderate
Kokubo 2007 (37)	PC (JPHC)	Japan	General population	40-59	40462 (52)	Soybeans	13	CVD, CHD, Stroke	Public	Moderate
Martinez- Gonzales 2011 (96)	PC (SUN)	Spain	University graduates from all Spain	>25	14129 (60)	Legumes	Ŋ	CVD, CHD	Public	Moderate
Miller 2017 (45)	PC (PURE)	18 countries in seven geographi- cal regions1	General population	35-70	135335 (58)	Legumes	7	CVD, CHD, Stroke	Mixed	Moderate
Mizrahi 2009 (35)	PC (FMC)	Finland	General population	40-74	3932 (48)	Legumes	24	Stroke	Public	Moderate
Nagura 2009 (38)	PC (JACC)	Japan	General population	40-79	59485 (58)	Soybeans	13	CVD, CHD, Stroke	Public	Moderate
Nouri 2021 (52)	PC (ICS)	Iran	General population	≥35	5432 (51)	Legumes	13	CVD	None	Serious
Papandreou 2019 (53)	prospective analysis from PREDIMED	Spain	Adults at high risk of CVD	55-80	7212 (57)	Legumes	9	CVD	Mixed	Moderate
Perez-Cornago 2021 (44)	PC (EPIC)	10 European countries2	General population	35-65	490311 (71)	Legumes	13	CHD	Mixed	Moderate
Stefler 2017 (97)	PC (HAPIEE)	Poland, Russia and Czech Republic	General population	45-69	19263 (55)	Legumes	7	CVD, CHD, Stroke	Public	Moderate

Table I. (Contim	(pen									
Article, reference	Study design (cohort name, if applicable)	Country	Subjects	Age at entry	Number analyzed (% women)	Exposure	Follow-up (y) / Duration (weeks	Outcomes s)	Funding	Risk of bias
Tong 2020 (43)	PC (EPIC)	9 European countries3	General population	35-65	418329 (67)	Legumes	13	Stroke	Public	Moderate
Yamasaki 2015 (39)	PC (JMS)	Japan	General population	mostly 40-69	I 1066 (61)	Soybeans	12	CVD	Private	Serious
Yu 2014 (98)	PC (SWHS, SMHS)	China	Urban population	40-70 (SWHS), 40-74 (SMHS)	I 22685 (55)	Legumes	IO (SWHS), 5 (SMHS)	CHD	Public	Moderate
Observational stu	ıdies, diabetes									
Bazzano 2008 (57)	PC (NHS)	USA	Female registered nurses	38-63	71346 (100)	Legumes	8	T2D	Public	Moderate
Becerra-Tomás 2018 (55)	prospective analysis from PREDIMED	Spain	Adults at high risk of CVD	55-80	3349 (62)	Legumes	4	T2D	Mixed	Moderate
Ericson 2013 (36)	PC (MDC)	Sweden	General population	45-74	27140 (61)	Legumes	12	T2D	Mixed	Moderate
Hodge 2004 (46)	PC (MCCS)	Australia	Australia-born and migrants from Southern Europe	40-69	36787 (unclear)	Legumes	4	T2D	Public	Moderate
Khalili- Moghadam 2018 (54)	PC (TLGS)	Iran	General population	20-69	2139 (55)	Legumes	6	T2D	Not reported	Serious
Liu 2004 (99)	PC (WHS)	NSA	Female health professionals	≥45	38018 (100)	Legumes	6	T2D	Not reported	Moderate
Meyer 2000 (100)	PC (IWHS)	NSA	Postmenopausal women	55-69	35988 (100)	Legumes	9	T2D	Public	Serious
O'Connor 2020 (56)	PC (ARIC)	USA	General population	45-65	I 1991 (56)	Legumes	22	T2D	Public	Moderate
Villegas 2008 (42)	PC (SWHS)	China	Female urban population	40-70	64191 (100)	Legumes and soybeans	Ŋ	T2D	Public	Moderate
Yan 2021 (40) Intervention studi	PC (JACC) ies	Japan	General population	40-79	21925 (62)	Soybeans	5	T2D	Public	Moderate
Abeysekara 2012 (58)	crossover trial	Canada	Healthy adults over 50 y, no hypolipidemic medications (n=20 had high cholesterol)	≥50	80 (арргох. 71)	Legumes	ω	risk factors	Mixed	Some concerns
Azadbakht 2007 (59)	crossover trial	Iran	Postmenopausal women with MetS	postmenopausal women	42 (100)	Soy-nut	8	risk factors	Public	Some concerns
Bakhtiari 2019 (60)	parallel trial	Iran	Postmenopausal women with MetS	60-70	25 (100)	Soy-nut	12	risk factors	Public	Low
Cobiac 1990 (101)	crossover trial	Australia	Mildly hypercholesterolemic men, healthy and no hypolip- idemic medications	29-65	20 (0)	Beans	4	risk factors	Private	Some concerns

Table I. (Continu	(per									
Article, reference	Study design (cohort name, if applicable)	Country	Subjects	Age at entry	Number analyzed (% women)	Exposure	Follow-up (y) / Duration (weeks)	Outcomes	Funding	Risk of bias
Doma 2021 (61)	crossover trial	Canada	Adults with elevated LDL- cholesterol, no hypolipid- emic medications	≥I8Y (mean±SD: 48±I4)	66 (approx. 56)	Beans	4	risk factors	Mixed	Low
Duane 1997 (65)	crossover trial	USA	Healthy men living on met- abolic ward for the purpose of the study	41-78	(0) 6	Legumes	9	risk factors	Private	Some concerns
Lin 1981 (62)	non-randomized trial	China	Adults with asymptomatic hyperlipidemia	27-66	Invervention n = 73, Control n = 169 (unclear)	Beans	12	risk factors	Not reported	Critical
Mackay 1992 (47)	crossover trial	New Zealand	Hypercholesterolaemic adults	28-66	39 (44)	Beans	6	risk factors	Mixed	Some concerns
Mizelman 2020 (67)	crossover trial	Canada	Healthy young athletes	young (mean±SD 22±6)	10 (50)	Legumes	4	risk factors	Private	Some concerns
Nestel 2004 (48)	crossover trial	Australia	Healthy adults, no hypolipid- emic medications	30-70	20 (50)	Chickpeas	6	risk factors	Private	Some concerns
Pittaway 2006 (49)	crossover trial	Australia	Healthy adults, no hypolipid- emic medications	30-70	47 (60)	Chickpeas	5-6	risk factors	Private	Some concerns
Pittaway 2007 (50)	crossover trial	Australia	Healthy adults, no hypolipid- emic medications	30-70	27 (63)	Chickpeas	S	risk factors	Private	Some concerns
Saraf-Bank 2016 (68)	crossover trial	Iran	Healthy first-degree relatives of patients with diabetes	50±7	26 (54)	Legumes	6	risk factors	Public	Some concerns
Tischmann 2022 (66)	crossover trial	the Netherlands	Healthy older adults	60-70	23 (52)	Soy-nut	16	risk factors	Private	Low
Winham 2007a (64)	crossover trial	USA	Healthy adults, fasting insulin level 15-50 µU/ml, no cholesterol-lowering medications	22-65	l 6 (56)	Pinto beans, black eyed peas	ω	risk factors	Mixed	Some concerns
Winham 2007b (63)	crossover trial	NSA	Hypercholesterolaemic adults	22-70	23 (57)	Beans	ω	risk factors	Private	Some concerns
Abbreviations: PC	prospective cohort	; FFQ, food frequenc;	y questionnaire; FR, food record							

Examination Survey; GCS, Golestan Cohort Study; HAPIEE, Health Alcohol and Psychosocial Factors in Eastern Europé; HPFS, Health Professionals Follow-Up Study; ICS, Isfahan cohort study; IWHS, Iowa Women's Health Study; JACC, Japan Collaborative Cohort Study; JMS, Jichi Medical School Cohort Study; JHC, Japan Public Health Center-based Prospective Study; MCCS, Melbourne Collaborative Cohort Study: MDC, Malmö Diet and Cancer Study: NHS, Nurses Health Study: PLSAW, Perth Longitudinal Study of Aging in Women; PREDIMED. Prevención con Dieta Mediterránea; PURE, Prospective Urban Rural I Miller 2017, 18 countries: Canada, Sweden, United Arab Emirates, Argentina, Brazil, Chile, Poland, Turkey, Malaysia, South Africa, China, Colombia, Iran, the occupied Palestinian territory, Bangladesh, India, Abbreviated cohort names: AHS, Adventist Health Study; ARIC, Atherosclerosis Risk in Communities, EPIC, European Prospective Investigation into Cancer and Nutrition; FMC, Finnish Mobile Clinic Health Epidemiology; SUN, Seguimiento Universidad de Navarra Study; SMHS, Shanghai Men's Health Study; SVHS, Shanghai Women's Health Study; TLGS, Tehran Lipid and Glucose Study; WHS, Women's health study. Pakistan, Zimbabwe.

8 (page number not for citation purpose) the included studies were considered not having, for the purpose of the present systematic review, an optimal control intervention; Mackay compared beans with low or high fibre oat bran (73) while Nestel and Pittaway compared chickpeas with a whole grain wheat diet (75–77).

Risk of bias in included studies

The summary RoB assessments of the included cohort studies and RCTs are shown in Figs. 2 and 3, respectively. Study level RoB assessments are shown in Supplementary Fig. 1 (cohort studies and the single non-randomized trial) and Supplementary Fig. 2 (RCTs, both crossover and parallel).

The overall RoB in cohort studies was regarded as moderate for 27 articles and serious for 4 articles. While some potential for confounding of the effect of exposure was expected in all studies, those not measuring energy intake (identified as a key confounder) were regarded as having serious RoB in domain 1. Some bias in domain 2 was expected for all studies since the start of follow-up was long after participants likely started consuming legumes. When outcome assessment was based on hospital registries, death certificates, or otherwise confirmed, the RoB for measurement of outcomes (domain 6) was regarded as low, while self-reported assessments that were partly confirmed were regarded as moderate RoB, and when possibly invalid, as serious RoB. The overall RoB in the non-randomized intervention was regarded as critical due to issues with confounding and missing data. The RoB for three RCTs was regarded as low and moderate for the remaining 12 RCTs. The lack of a washout period raised concerns (domain 1), as did low compliance (domain 2), lack of blinding of outcome assessors (domain 4) and lack of a published analysis plan, especially considering the private funding (domain 5).

Summary of findings from observational studies

Five individual observational studies (four regarded with moderate and one with serious RoB) reported significant associations between legume intake and CVD outcomes in fully adjusted models (see Supplementary Table 3). Four reported inverse associations: two between total legume intake and CVD incidence (40, 49), one between total legume intake and ischemic stroke (but not total stroke) (47) and one between soybean intake and CVD mortality (48). On the other hand, a prospective analysis from the PREDIMED trial reported an association between total legume intake and higher CVD mortality (50).

Out of the 12 studies reporting on overall CVD, nine provided sufficient data to be included in the meta-analysis (3 studies on incidence and 6 studies on mortality). The analysis resulted in a pooled estimate for high versus low legume intake of RR 0.95 (95% CI, 0.86, 1.06, $I^2 = 41\%$) for total (or 'overall') CVD. An assessment of







Fig. 3. Summary risk of bias per domain in RCTs (n = 15).

CVD mortality alone resulted in a pooled estimate of RR 1.03 (95% CI, 0.89, 1.20, $I^2 = 48\%$) (Fig. 4A). For CHD, 10 out of 11 studies provided sufficient data to be included in the high versus low legume intake meta-analysis (7 studies on incidence and 3 on mortality). The analysis resulted in a pooled estimate of RR 1.00 (95% CI, 0.95, 1.05, $I^2 = 0\%$) for total ('overall') CHD and an estimate of RR 0.99 (95% CI, 0.94, 1.05, $I^2 = 0\%$) for CHD incidence alone (Fig. 4B). For stroke, all nine studies retrieved (6 on incidence and 3 on mortality) were included in the high versus low meta-analysis; RR 0.98 (95% CI, 0.91, 1.05, $I^2 = 16\%$) for total ('overall') stroke and RR 0.99 (95% CI, 0.91, 1.05, $I^2 = 16\%$) for stroke incidence alone (Fig. 4C). No clear dose-response association was found for any of the outcomes.

Half of the studies on legume intake and T2D risk (four regarded with moderate and one with serious RoB) reported significant associations. Becerra-Tomás, Khalili-Moghadam, O'Connor and Villegas reported inverse associations between total legume intake and T2D (57, 60, 63, 64). Furthermore, lentils and soybeans were inversely associated with T2D risk in Becerra-Tomás and Villegas (64), respectively. In the opposite direction, Bazzano reported significant associations between legume intake and higher T2D (56). Nine out of 10 studies reporting on T2D provided sufficient data to be included in the meta-analysis, resulting in a pooled estimate for legume intake of RR 0.90 (95% CI, 0.77, 1.06, $I^2 = 88\%$) and no clear dose response association (Fig. 4D).

Separate assessment by legume type (overall legumes excluding soybeans vs. soybeans alone) resulted in pooled estimates for high versus low legume (excluding soybean) intake of RR 1.00 (95% CI, 0.95, 1.06, $f^2 = 0\%$) for overall CHD, RR 0.97 (95% CI, 0.89, 1.07, F = 35%) for overall stroke and RR 0.89 (95% CI, 0.74, 1.07, F = 89%) for T2D (Supplementary Fig. 3). There were no differences between studies based on funding source (not shown).

Summary of findings from intervention studies

As shown in Supplementary Table 4, 12 out of the 16 intervention studies reported some significant effects, all considered protective for the outcomes of interest. A significant lowering of TC and/or LDL-C was observed in 10 intervention studies (65–68, 70–72, 76, 79–81); the effect on TC was observed in 8 studies and LDL-C also in 8 studies. These studies included healthy middle-aged or older adults, some with high cholesterol, and post-menopausal women with MetS, in interventions on mixed legumes (\geq 120–150 g/day dry weight), soynut (\geq 30 g/day, beans (\geq 50 g/day cooked weight), chickpeas (140 g/day cooked weight) or pinto beans (130 g/day cooked weight). Significant increases in HDL-cholesterol was observed in studies on hypercholesterolemic adults consuming beans

(80 g/day cooked weight) and healthy young athletes consuming mixed legumes (156 g/day dry weight) (73, 74). Changes in insulin, glucose and HOMA-IR were only seen in RCTs from Iran on postmenopausal women with MetS consuming 30–35 g/day soynuts (67, 68). No effects were seen on blood pressure (67, 78).

All RCTs (crossover and parallel) were included in the meta-analysis, while the non-randomized trial was not (72). The summary effect sizes (Fig. 5) showed significantly decreased TC (-0.22 mmol/L, 95% CI, -0.32, -0.13, $I^2 = 75\%$), LDL-cholesterol (-0.19 mmol/L, 95% CI, -0.27, -0.11, $I^2 = 52\%$), fasting glucose (-0.19 mmol/L, 95% CI, -0.33, -0.05, $I^2 = 83\%$), and HOMA-IR (-0.30, 95% CI, -0.60, -0.00, $I^2 = 96\%$) with legume interventions compared with controls, but insignificant effects on HDL-C, TG, and insulin. Heterogeneity was high but decreased when separated by type of legumes (Supplementary Fig. 4). Especially, excluding the soynut trials removed nearly all heterogeneity. Excluding soynut trials, the intervention effect of legumes was -0.17 mmol/L (95% CI -0.25, -0.09, *I*² = 31%) for TC, -0.15 mmol/L (95% CI -0.21, -0.09, $I^2 = 5\%$) for LDL-C, -0.09 mmol/L (95% CI $-0.19, 0.01, I^2 = 50\%$) for glucose and 0.01 (95% CI -0.01, 0.03, $I^2 = 0\%$) for HOMA-IR. The effects of soynuts on TC, fasting glucose and insulin were significantly different from other interventions (P = 0.03 for TC, P = 0.003 for glucose and P < 0.001 for insulin). For glucose, the intervention effect was also larger in studies 6 weeks or longer than 4-5 weeks (data not shown). Subgroup analysis by RoB gave similar results for RCTs with some concerns and all studies (Supplementary Fig. 5). The effect sizes did not vary by funding source (not shown), except for larger effects in the two publicly funded RCTs using soynuts for glucose, insulin, and HOMA-IR.

Reporting bias

Based on inspection of funnel plots (Supplementary Fig. 6), and Egger's test (in meta-analyses with 10 or more comparators), we did not find evidence of publication bias in the form of small study-effects bias. Egger's test for legumes and CHD gave a *P*-value of 0.62, and for the risk factors the *P*-values ranged from 0.38 to 0.98, with the exception of HDL-C (P = 0.06) where one small trial found a large increase in females (n = 5).

Certainty in the evidence

Overall, the findings from this systematic review are mixed, with high heterogeneity and mostly moderate/ some concern for RoB (Table 2). Although it was not the purpose of this review, we consider the evidence strong in supporting the absences of any adverse effects of legume intake in relation to CVD or T2D. Nevertheless, meta-analyses of prospective cohort studies on either



Fig. 4. Legume consumption and risk of cardiometabolic endpoints in cohort studies, shown overall and separated by incidence and mortality: CVD (A), CHD (B), stroke (C), T2D (D). The upper figures show summary forest plots of pooled relative risk estimates between highest and lowest legume consumption categories and risk of the endpoints. The lower figures show linear (dashed line) and non-linear (solid line) dose-response associations (with 95% CI) between legume consumption and risk of the endpoints (total CVD, total CHD, total stroke, T2D), with 0 g/day as reference and vertical axes log transformed.



9 studies (same as in forest plot except Fung 2018) P for linearity = 0.4 (HR per 100 g = 0.97, 95 % CI: 0.89, 1,05) P for nonlinearity = 0.47

Fig. 4. Continued Legume consumption and risk of cardiometabolic endpoints in cohort studies, shown overall and separated by incidence and mortality: CVD (A), CHD (B), stroke (C), T2D (D). The upper figures show summary forest plots of pooled relative risk estimates between highest and lowest legume consumption categories and risk of the endpoints. The lower figures show linear (dashed line) and non-linear (solid line) dose-response associations (with 95% CI) between legume consumption and risk of the endpoints (total CVD, total CHD, total stroke, T2D), with 0 g/day as reference and vertical axes log transformed.



Fig. 4. Continued Legume consumption and risk of cardiometabolic endpoints in cohort studies, shown overall and separated by incidence and mortality: CVD (A), CHD (B), stroke (C), T2D (D). The upper figures show summary forest plots of pooled relative risk estimates between highest and lowest legume consumption categories and risk of the endpoints. The lower figures show linear (dashed line) and non-linear (solid line) dose-response associations (with 95% CI) between legume consumption and risk of the endpoints (total CVD, total CHD, total stroke, T2D), with 0 g/day as reference and vertical axes log transformed.

С





9 studies P for linearity = 0.54 (HR per 100 g = 0.95, 95 % CI 0.80, 1.12) P for nonlinearity = 0.25

Fig. 4. Continued Legume consumption and risk of cardiometabolic endpoints in cohort studies, shown overall and separated by incidence and mortality: CVD (A), CHD (B), stroke (C), T2D (D). The upper figures show summary forest plots of pooled relative risk estimates between highest and lowest legume consumption categories and risk of the endpoints. The lower figures show linear (dashed line) and non-linear (solid line) dose-response associations (with 95% CI) between legume consumption and risk of the endpoints (total CVD, total CHD, total stroke, T2D), with 0 g/day as reference and vertical axes log transformed.

total CVD, CHD, stroke or T2D showed null associations comparing higher with lower legume consumption and no dose–response relationship. This would indicate substantial effects unlikely but needs to be interpreted in the light of the generally low legume consumption reported in the cohorts. RCTs on approximately

~			Mean difference	Weight
Study	Intervention		95 % CI	(%)
Abeysekara, 2012	Legumes	←∎	-0.38 (-0.59, -0.17)	7.34
Duane, 1997	Legumes	← ■	-0.26 (-0.51, -0.01)	6.35
Mizelman, 2020 (female)	Legumes		0.00 (-0.16, 0.16)	8.73
Mizelman, 2020 (male)	Legumes	← ■	-0.20 (-0.51, 0.11)	5.09
Saraf-Bank, 2016	Legumes		-0.11 (-0.31, 0.09)	7.57
Cobiac, 1990	Beans		-0.03 (-0.35, 0.29)	4.99
Doma, 2021	Beans		-0.21 (-0.41, -0.01)	7.64
Mackay, 1992	Beans		0.00 (-0.24, 0.24)	6.61
Winham, 2007a	Beans		-0.26 (-0.49, -0.03)	6.93
Winham, 2007b	Beans		-0.13 (-0.45, 0.19)	5.02
Azadbakht, 2007	Soy nuts		-0.49 (-0.53, -0.45)	11.28
Bakhtiari, 2019	Soy nuts	←───	-0.62 (-0.97, -0.26)	4.36
Tischmann, 2022	Soy nuts		-0.17 (-0.34, -0.01)	8.52
Pittaway, 2006	Chickpeas	_	-0.22 (-0.35, -0.09)	9.56
Overall		\diamond	-0.22 (-0.32, -0.13)	
Heterogeneity: $\tau^2 = 0.02$, I	² = 75.48%, H ² = 4.08			
		54321 0 .1 .2 .3 mmol/l	.4 .5	

В

Δ

Study	Intervention		Mean difference 95 % Cl	Weight (%)
Abeysekara, 2012	Legumes	_	-0.23 (-0.43, -0.03)	8.27
Duane, 1997	Legumes	← ■	-0.31 (-0.58, -0.04)	5.74
Saraf-Bank, 2016	Legumes		-0.08 (-0.27, 0.10)	9.00
Cobiac, 1990	Beans		-0.02 (-0.26, 0.22)	6.61
Doma, 2021	Beans	_	-0.24 (-0.40, -0.08)	10.41
Mackay, 1992	Beans		0.05 (-0.15, 0.25)	8.12
Winham, 2007a	Beans	_	-0.13 (-0.34, 0.08)	7.88
Winham, 2007b	Beans		-0.09 (-0.39, 0.21)	5.01
Azadbakht, 2007	Soy nuts	←∎	-0.41 (-0.56, -0.27)	11.17
Bakhtiari, 2019	Soy nuts	<	-0.53 (-0.96, -0.10)	2.83
Tischmann, 2022	Soy nuts		-0.17 (-0.32, -0.02)	11.03
Pittaway, 2006	Chickpeas		-0.18 (-0.28, -0.08)	13.93
Overall		\diamond	-0.19 (-0.27, -0.11)	
Heterogeneity: τ^2 =	: 0.01, I ² = 51.62%, H ² = 2.07	54321 0 .1 .2 .3 .4	5	

Fig. 5. Effect of legume intervention versus control in RCTs. The figures show summary forest plots of pooled mean differences between intervention and control in cardiometabolic risk factors: total cholesterol, mmol/L (A), LDL-cholesterol, mmol/L (B), HDL-cholesterol, mmol/L (C), triglycerides, mmol/L (D), glucose, mmol/L (E), insulin, pmol/L (F), HOMA-IR (G).

>120–150 g legumes per day suggested a biological plausibility, with most, but not all studies reporting lowering of TC and LDL-C and meta-analyses additionally suggesting protective effects on glucose and HOMA-IR. Since legume interventions were suggested to have protective effects on blood lipids and glycemic markers, we did not consider the evidence strong enough to support a convincing absence of a causal relationship.

С				
Study	Intervention		Mean difference 95 % Cl	Weight (%)
				(7-7)
Abeysekara, 2012	Legumes		-0.07 (-0.21, 0.07	6.02
Duane, 1997	Legumes	₽	-0.26 (-0.29, -0.22) 10.17
Mizelman, 2020, female	Legumes		→ 0.70 (0.21, 1.19	0.98
Mizelman, 2020, male	Legumes	← ■	-0.20 (-0.62, 0.22) 1.27
Saraf-Bank, 2016	Legumes		0.02 (-0.06, 0.10	8.39
Cobiac, 1990	Beans	_	-0.04 (-0.13, 0.05) 8.12
Doma, 2021	Beans		-0.05 (-0.12, 0.02) 8.78
Mackay, 1992	Beans		-0.01 (-0.09, 0.07	8.58
Winham, 2007a	Beans	#	0.00 (-0.14, 0.14	6.06
Winham, 2007b	Beans	#	0.00 (-0.12, 0.12	6.62
Azadbakht, 2007	Soy nuts	-	0.00 (-0.03, 0.03) 10.34
Bakhtiari, 2019	Soy nuts	— —	-0.07 (-0.17, 0.04) 7.37
Tischmann, 2022	Soy nuts	-	-0.03 (-0.07, 0.01	9.91
Pittaway, 2006	Chickpeas	_	-0.02 (-0.12, 0.08) 7.40
Overall		\diamond	-0.04 (-0.09, 0.01)
Heterogeneity: $\tau^2 = 0.01$,	$I^2 = 84.16\%, H^2 = 6.31$			
		54521 0 .1 .2 .3 .4 mmol/l		

D

			Mean difference	Weight
Study	Intervention		95 % CI	(%)
Abeysekara, 2012	Legumes		0.09 (-0.12, 0.30)	2.04
Duane, 1997	Legumes		0.14 (-0.09, 0.36)	1.64
Saraf-Bank, 2016	Legumes	+•	0.08 (-0.17, 0.32)	1.44
Cobiac, 1990	Beans	_ -	0.01 (-0.13, 0.15)	4.30
Doma, 2021	Beans		0.05 (-0.03, 0.12)	13.87
Mackay, 1992	Beans		0.03 (-0.13, 0.19)	3.21
Winham, 2007a	Beans	<	-0.32 (-0.60, -0.03)	1.08
Winham, 2007b	Beans		-0.15 (-0.45, 0.16)	0.91
Azadbakht, 2007	Soy nuts	+	-0.01 (-0.04, 0.02)	61.25
Bakhtiari, 2019	Soy nuts		-0.10 (-0.40, 0.21)	0.92
Tischmann, 2022	Soy nuts	_ -	0.03 (-0.07, 0.13)	7.92
Pittaway, 2006	Chickpeas		-0.04 (-0.29, 0.21)	1.41
Overall		\diamond	0.00 (-0.03, 0.03)	
Heterogeneity: τ^2 =	$0.00, ^2 = 3.76\%, H^2 = 1.04$			
		54321 0 .1 .2 .3 .4	ר .5	
		ΠΠΟΙ/Ι		

Fig. 5. Continued Effect of legume interventionversus control in RCTs. The figures show summary forest plots of pooled mean differences between intervention and control in cardiometabolic risk factors: total cholesterol, mmol/L (A), LDL-cholesterol, mmol/L (B), HDL-cholesterol, mmol/L (C), triglycerides, mmol/L (D), glucose, mmol/L (E), insulin, pmol/L (F), HOMA-IR (G).

-					Mean difference	Weight
Study	Intervention				95 % CI	(%)
Abeysekara, 2012	Legumes				→ 0.32 (-0.38, 1.02)	3.20
Saraf-Bank, 2016	Legumes	←	-		→-0.11 (-0.73, 0.50)	3.88
Cobiac, 1990	Beans			_	-0.03 (-0.12, 0.06)	14.24
Doma, 2021	Beans		_		0.03 (-0.08, 0.14)	14.00
Winham, 2007a	Beans				-0.28 (-0.47, -0.09)	11.76
Winham, 2007b	Beans	←∎			-0.22 (-0.53, 0.09)	8.77
Azadbakht, 2007	Soy nuts				-0.50 (-0.59, -0.41)	14.36
Bakhtiari, 2019	Soy nuts	←	_		-0.51 (-0.90, -0.12)	7.01
Nestel, 2004	Chickpeas		 		-0.20 (-0.39, -0.01)	12.02
Pittaway, 2007	Chickpeas				-0.07 (-0.30, 0.16)	10.76
Overall		<	>		-0.19 (-0.33, -0.05)	
Heterogeneity: τ^2 =	= 0.03, I ² = 82.77%, H ² = 5.80					
		543	21 C) .1 .2 .3 .4 ol/l	.5	

F

							Mean difference	e	Weight
Study	Intervention						95 % CI		(%)
Abeysekara, 2012	Legumes						— 1.50 (-25.80, 28	8.80)	2.71
Mizelman, 2020 female	Legumes				-		7.80 (-3.34, 18	8.94)	10.35
Mizelman, 2020 male	Legumes			+		_	-0.60 (-19.99, 18	8.79)	4.83
Doma, 2021	Beans						-0.40 (-1.20, 0	0.40)	23.58
Winham, 2007a	Beans			•			-6.00 (-27.03, 15	5.03)	4.23
Winham, 2007b	Beans	←			_		-12.00 (-33.03, 9	9.03)	4.23
Azadbakht, 2007	Soy nuts						-8.40 (-9.35, -7	7.45)	23.52
Bakhtiari, 2019	Soy nuts	←	-	-			-18.00 (-32.08, -3	3.92)	7.74
Nestel, 2004	Chickpeas				_		-1.80 (-12.91,	9.31)	10.39
Pittaway, 2007	Chickpeas		_	╺┼┼═╸		-	3.24 (-9.98, 16	5.46)	8.42
Overall			<	\Rightarrow			-3.32 (-8.09, 2	1.45)	
Heterogeneity: $\tau^2 = 25.0$	$0, I^2 = 93.52\%, H^2 = 15.44$								
		-30 -2	010	- 	10	20	30		
		50 -Z	0 -10	pmol/l	10	20	50		

Fig. 5. Continued Effect of legume interventionversus control in RCTs. The figures show summary forest plots of pooled mean differences between intervention and control in cardiometabolic risk factors: total cholesterol, mmol/L (A), LDL-cholesterol, mmol/L (B), HDL-cholesterol, mmol/L (C), triglycerides, mmol/L (D), glucose, mmol/L (E), insulin, pmol/L (F), HOMA-IR (G).

However, the direction of effect in observational studies was not considered consistent enough to be suggestive of an association (as in the grading *limited* – *suggestive*). Therefore, we judged the evidence for associations between legume intake and CVD or T2D as *limited* – *no conclusion*.

Discussion

This comprehensive systematic review and meta-analysis of both observational and intervention studies is, to our knowledge, the most recent update on the evidence for associations between legume intake and CVD and T2D and their risk factors. While legume RCTs were found 0

G							Moo	n difforanco	Woight
Study	Intervention						Wea	95 % Cl	(%)
Doma, 2021	Beans						0.01 (-0.01, 0.03)	24.17
Nestel, 2004	Beans					-	-0.10 (-0.58, 0.38)	14.81
Pittaway, 2007	Beans				-		0.10 (-0.49, 0.69)	12.40
Winham, 2007a	Beans						-0.30 (-1.28, 0.68)	6.70
Winham, 2007b	Beans	-	-				-0.55 (-1.54, 0.44)	6.58
Azadbakht, 2007	Soy nuts		∎				-0.60 (-0.67, -0.53)	23.86
Bakhtiari, 2019	Soy nuts	<∎		_			-0.90 (-1.54, -0.26)	11.48
Overall			<	\geq			-0.30 (-0.60, -0.00)	
Heterogeneity: τ ²	= 0.10, I ² = 95.84%, H ² = 24.04	-1	5)	.5	 1		

Fig. 5. Continued Effect of legume intervention versus control in RCTs. The figures show summary forest plots of pooled mean differences between intervention and control in cardiometabolic risk factors: total cholesterol, mmol/L (A), LDL-cholesterol, mmol/L (B), HDL-cholesterol, mmol/L (C), triglycerides, mmol/L (D), glucose, mmol/L (E), insulin, pmol/L (F), HOMA-IR (G).

having favorable effects on blood lipids and glycemic markers, which are risk factors for CVD and T2D (1), this systematic review found little support, but also little opposition, for recommending dietary inclusion of legumes for the purpose of CVD and T2D prevention for healthy general populations in the Nordic and Baltic countries. Data were too limited to draw conclusions about blood pressure or specific types of legumes. The evidence for associations between legume intake and CVD or T2D was considered *limited – no conclusion.*

Comparison with other reviews and discussion on heterogeneity Previous systematic reviews and meta-analyses on legumes have similarly found conflicting evidence, often considered of low or limited strength. Some, but not all systematic reviews on observational studies have found evidence for an association of overall legume intake with decreased risk of CVD and/or CHD in high versus low comparison of included studies or dose-response analysis (28, 29, 82-85). We are not aware of systematic reviews reporting evidence for an association between legume intake and stroke (28, 29, 82, 86, 87). Previous systematic reviews have also assessed the evidence for specific types of legumes, notably finding no or some evidence for soy intake and CVD (88-90). Weak or null associations between legume intake and T2D have been found (91) and recently, intakes of tofu, soy protein, and soy isoflavones, but not total legumes or total soy, were found to be associated with lower T2D incidence (92). This is interesting in the light of our observed differences between effects of soynut and other legume interventions in crossover trials. Meta-analyses on RCTs have previously found suggestions of protective effects of diets rich in legumes on TC

and LDL-cholesterol (93, 94). Another recent systematic review without meta-analyses on RCTs of legume interventions found evidence for improvements in blood lipid profile, blood pressure, inflammation biomarkers, and other measures resulting from consumption of approximately 150 g/day of legumes, similar to our finding (95), while only significant effects of interventions on glycemic control were found for individuals with established T2D in another review of RCTs including individuals with or without T2D or prediabetes (96).

The different findings in the current and some previous systematic reviews and meta-analyses may depend on many factors. Combining incidence and mortality outcomes, combining consumption of soy and other legumes, and pooling findings for men and women when reported separately may, for example, potentially result in heterogeneity in findings where null findings for some may mask findings for others. For example, in a meta-analysis from 2017, heterogeneity was considered to be due to null results of stroke, affecting estimates for total CVD (82). In our analyses, excluding soynut trials removed nearly all statistical heterogeneity in intervention studies, while separating by type of legumes (overall legumes vs. soy) did not affect heterogeneity in observational studies.

The setting, amounts of legumes consumed and broader dietary context of legume consumption are other important factors to consider and need further investigation. A recent meta-analysis found legume intake to be associated with higher T2D in Europe, mainly driven by studies from Germany, UK, and Sweden; but there is no evidence of associations between legume intake and T2D in the Americas, Eastern Mediterranean, and Western Pacific (91). Interestingly, in our analyses on

Table 2. Summary of findings

Outcome	Study type	Studies total (n included in meta-analysis)	Exposure (n studies)	Association/ effect (n studies)	Meta-analysis results	Heterogeneity (I2)	Risk of bias (n studies)	Evidence assessment
CVD	Observational	12 (9)	Legumes (8) Soybeans (4)	$ \begin{array}{c} \uparrow \ I \\ \downarrow \ 3 \\ \leftrightarrow \ 8 \end{array} $	0.95 (0.86, 1.06) 1 No clear dose-response	40.70%	Moderate: 10 Serious: 2	Mixed findings, in favor of null association
CHD	Observational	11 (10)	Legumes (9) Soybeans (2)	↑ 0 ↓ 0 ↔ I I	1.00 (0.95, 1.05) 1 No clear dose-response	0.00%	Moderate: 10 Serious: I	In favor of null association
Stroke	Observational	9 (9)	Legumes (7) Soybeans (2)	↑ 0 ↓ I ↔ 8	0.98 (0.91, 1.05) 1 No clear dose-response	16.03%	Moderate: 9	In favor of null association
T2D	Observational	10 (9)	Legumes (9) Soybeans (1)	↑ I ↓ 4 ↔ 5	0.90 (0.77, 1.06) 1 No clear dose-response	88.37%	Moderate: 8 Serious: 2	Mixed findings, in favor of null association
Total cholesterol	Interventions	14 (13)	Mixed (4) Beans (5) Soy-nuts (3) Chickpeas (1) Pinto/black eyed (1)	$ \begin{array}{c} \uparrow 0 \\ \downarrow 8 \\ \leftrightarrow 6 \end{array} $	-0.22 (-0.32, -0.13) mmol/L 2	75.48%	Low: 3 Some concern: 10 Critical: 1	Suggestions of a protec- tive effect
LDL- cholesterol	Interventions	12 (12)	Mixed (3) Beans (4) Soy-nuts (3) Chickpeas (1) Pinto/black eyed (1)	$ \begin{array}{c} \uparrow 0 \\ \downarrow 8 \\ \leftrightarrow 4 \end{array} $	-0.19 (-0.27, -0.11) mmol/L 2	51.62%	Low: 3 Some concern: 9	Suggestions of a protec- tive effect
HDL- cho- lesterol	Interventions	13 (13)	Mixed (4) Beans (4) Soy-nuts (3) Chickpeas (1) Pinto/black eyed (1)	$ \begin{array}{c} \uparrow 2 \\ \downarrow 0 \\ \leftrightarrow \parallel 1 \end{array} $	-0.04 (-0.09, 0.01) mmol/L 2	84.16%	Low: 3 Some concern: 10	In favor of no effect
Triglycerides	Interventions	13 (12)	Mixed (3) Beans (5) Soy-nuts (3) Chickpeas (1) Pinto/black eyed (1)	↑ 0 ↓ 0 ↔ 3	0.00 (-0.03, 0.03) mmol/L 2	3.76%	Low: 3 Some concern: 9 Critical: I	In favor of no effect
Glucose	Interventions	10 (10)	Mixed (2) Beans (3) Soy-nuts (2) Chickpeas (2) Pinto/black eyed (1)	↑ 0 ↓ I ↔ 9	-0.19 (-0.33, -0.05) mmol/L 2	82.77%	Low: 2 Some concern: 8	Mixed find- ings, some suggestions of a protec- tive effect
Insulin	Interventions	9 (9)	Mixed (2) Beans (2) Soy-nuts (2) Chickpeas (2) Pinto/black eyed (1)	$ \begin{array}{c} \uparrow 0 \\ \downarrow I \\ \leftrightarrow 8 \end{array} $	-3.32 (-8.09, I.45) pmol/L 2	93.52%	Low: 2 Some concern: 7	Mixed findings, no conclusion
HOMA-IR	Interventions	7 (7)	Mixed (0) Beans (2) Soy-nuts (2) Chickpeas (2) Pinto/black eyed (1)	$ \begin{array}{c} \uparrow 0 \\ \downarrow 2 \\ \leftrightarrow 5 \end{array} $	-0.30 (-0.60, -0.00) 2	95.84%	Low: 2 Some concern: 7	Mixed find- ings, some suggestions of a protec- tive effect

Table 2. (Continued)

Outcome	Study type	Studies total (n included in meta-analysis)	Exposure (n studies)	Association/ effect (n studies)	Meta-analysis results	Heterogeneity (I2)	Risk of bias (n studies)	Evidence assessment
Systolic blood pressure	Interventions	2 (0)	Mixed (I) Soy-nuts (I)	$ \begin{array}{c} \uparrow 0 \\ \downarrow 0 \\ \leftrightarrow 2 \end{array} $	-	-	Some concern (2)	No conclusion
Diastolic blood pressure	Interventions	2 (0)	Mixed (1) Soy-nuts (1)	$ \begin{array}{c} \uparrow \ 0 \\ \downarrow \ 0 \\ \leftrightarrow 2 \end{array} $	-	-	Some concern (2)	No conclusion

Abbreviations: CHD, coronary heart disease; CVD, cardiovascular disease; HOMA-IR, homeostatic model assessment for insulin resistance; T2D, type 2 diabetes;

IRR (95% CI), comparing high vs. low consumption

2Mean difference (95% CI), comparing intervention vs. control

legume consumption and T2D, the studies with relative risks >1.10 were from the USA, while the studies with RR ≤ 0.65 were from Iran, China and Spain (a population consuming a Mediterranean-style diet). Residual confounding may be one source of this difference, for example, different overall dietary patterns and cooking methods of legumes (91), confounding by other components in the diet [e.g. consumption of legumes as parts of meat-based meals in Western Europe (97) and possibly the USA versus consumption of legumes with olive oil in the Mediterranean region (98)].

Strengths and limitations

The main strength of this review is the established process for undertaking robust systematic reviews. A priori, the NNR 2022 Committee established criteria for the prioritization and selection of systematic review topics. Prior to undertaking the review, we developed and registered a detailed protocol to enhance transparency in the review process (12, 13). We searched four leading electronic databases to identify relevant studies on the review topic, these databases cover most of the literature in medicine and public health. The search was updated before publication, to be as up to date as possible. In addition to this comprehensive database search, we screened reference lists from included articles and relevant systematic reviews for additional potentially eligible studies. We included nine articles from this additional search, which may seem high, but we believe that it can all be explained by the exposure characterization, as legumes were often explored as a subgroup of vegetables, thus not mentioned in title/abstract. After our thorough article searching, we consider it unlikely that we missed any relevant literature, although it cannot be excluded. The review processes were rigorously implemented, with independent assessments taken at each stage, including literature screening and data extraction.

All observational studies were classified as having either moderate or serious RoB. The RCTs were classified as having low or some concerns for RoB, with the exception for the non-randomized trial. The RoB assessment tool we used for cohort studies is more appropriate for evaluating causality than commonly used summary scorebased quality appraisal tools, and since it partly assesses whether a cohort study mimics a RCT, it is very difficult for a study to get a low RoB. That being said, most of the included cohort studies were prone to some of the limitations inherent in many observational epidemiologic studies, including risk of residual confounding and risk associated with selection of participants into the study and drop-outs. We found in general no significant differences between studies with different sources of funding. However, it should be recognized that almost all RCTs had private or mixed private/public funding.

Since another systematic review by the same authors examined the evidence for whether replacing animal protein with plant protein reduces risk of CVD and T2D, substitution effects were not considered in the current review. Therefore, the comparison in observational studies was done on different exposure amounts, which sometimes needed to be estimated and are in general sensitive to risk of misclassification of exposure status and possibilities of change in exposure status. Findings from RCTs were mostly based on consumption of >120 g/day of legumes or specific types of legumes, which is very high considering the low consumption seen in the Nordic countries (9, 10).

Conclusion and public health relevance

Overall, the findings from this systematic review and meta-analysis were mixed, and evidence from observational studies, generally with low legume consumption, was suggestive of null associations. However, protective effects seen in RCTs on established risk factors for CVD and T2D lend some support for recommending legume consumption as part of a diverse and healthy dietary pattern for CVD and T2D prevention in healthy general populations in the Nordic and Baltic countries. Even though the effect sizes seemed clinically small, they may still be relevant in a life-course perspective on the population level (99). It should be noticed that the amounts of legumes were commonly higher in RCTs (>120–150 g/day legumes) than the mean in the highest intake category in cohort studies. Although the current intake in some Western countries may possibly be higher than what was commonly consumed in the cohort studies included in this review, we believe that intervention trials exploring effects of lower consumption levels on cardiometabolic biomarkers would be of public health relevance. Although the evidence for associations between legume intake and CVD or T2D was graded as *limited* – *no conclusion*, we consider the current results will inform the updated recommendations in the 2022 edition of NNR, which was the primary purpose of the review.

Acknowledgments

The authors would like to thank Hilde Strømme, Academic Librarian at the University of Oslo for her invaluable assistance with the literature searches, and the university librarians at Karolinska Institutet for peer reviewing the search strategy.

Conflict of interest and funding

The authors declare no potential conflict of interest. Partial funding was received from the Nordic Council of Ministers and governmental food and health authorities of Norway, Finland, Sweden, Denmark, and Iceland.

Registration

PROSPERO registration number CRD42021250084.

References

- GBD Risk Factors Collaborators. Global burden of 87 risk factors in 204 countries and territories, 1990–2019: a systematic analysis for the Global Burden of Disease Study 2019. Lancet 2020; 396(10258): 1223–49. doi: 10.1016/S0140-6736(20)30752-2
- Moreno-Valdespino CA, Luna-Vital D, Camacho-Ruiz RM, Mojica L. Bioactive proteins and phytochemicals from legumes: mechanisms of action preventing obesity and type-2 diabetes. Food Res Int 2020; 130: 108905. doi: 10.1016/j.foodres.2019.108905
- Rodríguez L, Mendez D, Montecino H, Carrasco B, Arevalo B, Palomo I, et al. Role of phaseolus vulgaris L. in the prevention of cardiovascular diseases-cardioprotective potential of bioactive compounds. Plants (Basel) 2022; 11(2): 186. doi: 10.3390/ plants11020186
- Bouchenak M, Lamri-Senhadji M. Nutritional quality of legumes, and their role in cardiometabolic risk prevention: a review. J Med Food 2013; 16(3): 185–98. doi: 10.1089/jmf.2011.0238
- Messina V. Nutritional and health benefits of dried beans. Am J Clin Nutr 2014; 100(Suppl 1): 437s–42s. doi: 10.3945/ ajcn.113.071472
- Willett W, Rockström J, Loken B, Springmann M, Lang T, Vermeulen S, et al. Food in the Anthropocene: the EAT-Lancet Commission on healthy diets from sustainable food systems. Lancet 2019; 393(10170): 447–92. doi: 10.1016/ S0140-6736(18)31788-4

- Semba RD, Ramsing R, Rahman N, Kraemer K, Bloem MW. Legumes as a sustainable source of protein in human diets. Global Food Security 2021; 28: 100520. doi: 10.1016/j.gfs.2021.100520
- Springmann M, Mozaffarian D, Rosenzweig C, Micha R. Chapter 02 What we eat matters: Health and environmental impacts of diets worldwide. 2021 Global Nutrition Report: The state of global nutrition. Bristol, UK: Development Initiatives; 2021.
- Lemming EW, Pitsi T. The Nordic Nutrition Recommendations 2022 – food consumption and nutrient intake in the adult population of the Nordic and Baltic countries. Food Nutr Res 2022; 66: 8572. doi: 10.29219/fnr.v66.8572
- Høyer A, Christensen JJ, Arnesen EK, Andersen R, Eneroth H, Erkkola M, et al. The Nordic Nutrition Recommendations 2022

 prioritisation of topics for de novo systematic reviews. Food Nutr Res 2021; 65: 7828. doi: 10.29219/fnr.v65.7828
- Arnesen EK, Christensen JJ, Andersen R, Eneroth H, Erkkola M, Høyer A, et al. The Nordic Nutrition Recommendations 2022 – handbook for qualified systematic reviews. Food Nutr Res 2020; 64: 4404. doi: 10.29219/fnr.v64.4404
- Arnesen EK, Christensen JJ, Andersen R, Eneroth H, Erkkola M, Høyer A, et al. The Nordic Nutrition Recommendations 2022 – structure and rationale of qualified systematic reviews. Food Nutr Res 2020; 64: 4403. doi: 10.29219/fnr.v64.4403
- Page MJ, McKenzie JE, Bossuyt PM, Boutron I, Hoffmann TC, Mulrow CD, et al. The PRISMA 2020 statement: an updated guideline for reporting systematic reviews. BMJ 2021; 372: n71. doi: 10.1136/bmj.n71
- Page MJ, Moher D, Bossuyt PM, Boutron I, Hoffmann TC, Mulrow CD, et al. PRISMA 2020 explanation and elaboration: updated guidance and exemplars for reporting systematic reviews. BMJ 2021; 372: n160. doi: 10.1136/bmj.n160
- Christensen JJ, Arnesen EK, Andersen R, Eneroth H, Erkkola M, Høyer A, et al. The Nordic Nutrition Recommendations 2022 – principles and methodologies. Food Nutr Res 2020; 64: 4402. doi: 10.29219/fnr.v64.4402
- Sterne JAC, Savović J, Page MJ, Elbers RG, Blencowe NS, Boutron I, et al. RoB 2: a revised tool for assessing risk of bias in randomised trials. BMJ 2019; 366: 14898. doi: 10.1136/bmj. 14898
- Higgins JPT, Li T, Sterne J on behalf of the RoB 2 working group on crossover trials. Revised Cochrane risk of bias tool for randomized trials (RoB 2) – Additional considerations for crossover trials 2021. Available from: https://www.riskofbias.info/welcome/ rob-2-0-tool/rob-2-for-crossover-trials [cited 12 September 2022].
- Sterne JA, Hernán MA, Reeves BC, Savović J, Berkman ND, Viswanathan M, et al. ROBINS-I: a tool for assessing risk of bias in non-randomised studies of interventions. BMJ 2016; 355: i4919. doi: 10.1136/bmj.i4919
- SDA Nutrition Evidence Systematic Review. Risk of Bias for Nutrition Observational Studies (RoB-NObs) tool 2019. Available from: https://nesr.usda.gov/sites/default/files/2019-07/ RiskOfBiasForNutritionObservationalStudies-RoB-NObs.pdf
- McGuinness LA, Higgins JPT. Risk-of-bias VISualization (robvis): an R package and Shiny web app for visualizing riskof-bias assessments. Res Synth Methods 2021; 12: 55–61. doi: 10.1002/jrsm.1411
- AHRQ Methods for Effective Health Care. Methods guide for effectiveness and comparative effectiveness reviews. Rockville, MD: Agency for Healthcare Research and Quality (US); 2008.

- 23. Higgins JP, Thomas J, Chandler J, Cumpston M, Li T, Page MJ, et al. eds. Cochrane Handbook for Systematic Reviews of Interventions version 6.3. Cochrane, 2022. Available from: www. training.cochrane.org/handbook [cited 12 September 2022].
- VanderWeele TJ. On a square-root transformation of the odds ratio for a common outcome. Epidemiology 2017; 28(6): e58–60. doi: 10.1097/EDE.00000000000733
- Greenland S. Quantitative methods in the review of epidemiologic literature. Epidemiol Rev 1987; 9: 1–30. doi: 10.1093/ oxfordjournals.epirev.a036298
- Marinangeli CPF, Curran J, Barr SI, Slavin J, Puri S, Swaminathan S, et al. Enhancing nutrition with pulses: defining a recommended serving size for adults. Nutr Rev 2017; 75(12): 990–1006. doi: 10.1093/nutrit/nux058
- Afshin A, Micha R, Khatibzadeh S, Mozaffarian D. Consumption of nuts and legumes and risk of incident ischemic heart disease, stroke, and diabetes: a systematic review and meta-analysis. Am J Clin Nutr 2014; 100(1): 278–88. doi: 10.3945/ajcn.113.076901
- Bechthold A, Boeing H, Schwedhelm C, Hoffmann G, Knüppel S, Iqbal K, et al. Food groups and risk of coronary heart disease, stroke and heart failure: a systematic review and dose-response meta-analysis of prospective studies. Crit Rev Food Sci Nutr 2019; 59(7): 1071–90. doi: 10.1080/10408398.2017.1392288
- 29. Viguiliouk E, Glenn AJ, Nishi SK, Chiavaroli L, Seider M, Khan T, et al. Associations between dietary pulses alone or with other legumes and cardiometabolic disease outcomes: an umbrella review and updated systematic review and meta-analysis of prospective cohort studies. Adv Nutr 2019; 10(Suppl_4): S308–s19. doi: 10.1093/advances/nmz113
- Crippa A, Discacciati A, Bottai M, Spiegelman D, Orsini N. One-stage dose-response meta-analysis for aggregated data. Stat Methods Med Res 2019; 28(5): 1579–96. doi: 10.1177/0962280218773122
- Orsini N, Greenland S. A procedure to tabulate and plot results after flexible modeling of a quantitative covariate. Stata J 2011; 11(1): 1–29. doi: 10.1177/1536867X1101100101
- 32. Higgins JP, Eldridge S, Li T. Including variants on randomized trials. In: Higgins JPT, Thomas J, Chandler J, Cumpston M, Li T, Page MJ, et al. eds. Cochrane handbook for systematic reviews of interventions version 6.3. Cochrane; 2022. Available from www.training.cochrane.org/handbook [cited 12 September 2022].
- 33. Balk EM, Earley A, Patel K, Trikalinos TA, Dahabreh IJ. AHRQ methods for effective health care. Empirical assessment of within-arm correlation imputation in trials of continuous outcomes. Rockville, MD: Agency for Healthcare Research and Quality (US); 2012.
- 34. Page MJ, Higgins JP, Sterne JA. Assessing risk of bias due to missing results in a synthesis. In: Higgins JPT, Thomas J, Chandler J, Cumpston M, Li T, Page MJ, et al. eds. Cochrane Handbook for systematic reviews of interventions version 6.3. Cochrane; 2022. Available from: www.training.cochrane.org/handbook [cited 12 September 2022].
- 35. Bernstein AM, Pan A, Rexrode KM, Stampfer M, Hu FB, Mozaffarian D, et al. Dietary protein sources and the risk of stroke in men and women. Stroke 2012; 43(3): 637–44. doi: 10.1161/STROKEAHA.111.633404
- 36. Blekkenhorst LC, Bondonno CP, Lewis JR, Devine A, Zhu K, Lim WH, et al. Cruciferous and allium vegetable intakes are inversely associated with 15-year atherosclerotic vascular disease deaths in older adult women. J Am Heart Assoc 2017; 6(10): e006558. doi: 10.1161/JAHA.117.006558

- 37. Farvid MS, Malekshah AF, Pourshams A, Poustchi H, Sepanlou SG, Sharafkhah M, et al. Dietary protein sources and all-cause and cause-specific mortality: the Golestan cohort study in Iran. Am J Prev Med 2017; 52(2): 237–48. doi: 10.1016/j. amepre.2016.10.041
- Fraser GE, Sabaté J, Beeson WL, Strahan TM. A possible protective effect of nut consumption on risk of coronary heart disease. The Adventist Health Study. Arch Intern Med 1992; 152(7): 1416–24. doi: 10.1001/archinte.1992.00400190054010
- 39. Fung TT, Isanaka S, Hu FB, Willett WC. International food group-based diet quality and risk of coronary heart disease in men and women. Am J Clin Nutr 2018; 107(1): 120–9. doi: 10.1093/ajcn/nqx015
- Golzarand M, Mirmiran P, Azizi F. Adherence to the MIND diet and the risk of cardiovascular disease in adults: a cohort study. Food Funct 2022; 13(3): 1651–8. doi: 10.1039/D1FO02069B
- 41. Haring B, Gronroos N, Nettleton JA, von Ballmoos MC, Selvin E, Alonso A. Dietary protein intake and coronary heart disease in a large community based cohort: results from the Atherosclerosis Risk in Communities (ARIC) study [corrected]. PLoS One 2014; 9(10): e109552. doi: 10.1371/journal. pone.0109552
- 42. Haring B, Misialek JR, Rebholz CM, Petruski-Ivleva N, Gottesman RF, Mosley TH, et al. Association of dietary protein consumption with incident silent cerebral infarcts and stroke: the atherosclerosis risk in communities (ARIC) study. Stroke 2015; 46(12):3 443–50. doi: 10.1161/STROKEAHA.115.010693
- 43. Im J, Park K. Association between soy food and dietary soy isoflavone intake and the risk of cardiovascular disease in women: a prospective cohort study in Korea. Nutrients 2021; 13(5): 1407. doi: 10.3390/nu13051407
- 44. Kokubo Y, Iso H, Ishihara J, Okada K, Inoue M, Tsugane S. Association of dietary intake of soy, beans, and isoflavones with risk of cerebral and myocardial infarctions in Japanese populations: the Japan Public Health Center-based (JPHC) study cohort I. Circulation 2007; 116(22): 2553–62. doi: 10.1161/ CIRCULATIONAHA.106.683755
- 45. Martínez-González MA, García-López M, Bes-Rastrollo M, Toledo E, Martínez-Lapiscina EH, Delgado-Rodriguez M, et al. Mediterranean diet and the incidence of cardiovascular disease: a Spanish cohort. Nutr Met Cardiovasc Dis 2011; 21(4): 237–44. doi: 10.1016/j.numecd.2009.10.005
- 46. Miller V, Mente A, Dehghan M, Rangarajan S, Zhang X, Swaminathan S, et al. Fruit, vegetable, and legume intake, and cardiovascular disease and deaths in 18 countries (PURE): a prospective cohort study. Lancet 2017; 390(10107): 2037–49. doi: 10.1016/S0140-6736(17)32253-5
- 47. Mizrahi A, Knekt P, Montonen J, Laaksonen MA, Heliövaara M, Järvinen R. Plant foods and the risk of cerebrovascular diseases: a potential protection of fruit consumption. Br J Nutr 2009; 102(7): 1075–83. doi: 10.1017/S0007114509359097
- Nagura J, Iso H, Watanabe Y, Maruyama K, Date C, Toyoshima H, et al. Fruit, vegetable and bean intake and mortality from cardiovascular disease among Japanese men and women: the JACC Study. Br J Nutr 2009; 102(2): 285–92. doi: 10.1017/S0007114508143586
- 49. Nouri F, Haghighatdoost F, Mohammadifard N, Mansourian M, Sadeghi M, Roohafza H, et al. The longitudinal association between soybean and non-soybean legumes intakes and risk of cardiovascular disease: Isfahan cohort study. Br Food J 2021; 123(8): 2864–79. doi: 10.1108/BFJ-08-2020-0699
- 50. Papandreou C, Becerra-Tomás N, Bulló M, Martínez-González M, Corella D, Estruch R, et al. Legume consumption and

risk of all-cause, cardiovascular, and cancer mortality in the PREDIMED study. Clin Nutr 2019; 38(1): 348–56. doi: 10.1016/j.clnu.2017.12.019

- 51. Perez-Cornago A, Crowe FL, Appleby PN, Bradbury KE, Wood AM, Jakobsen MU, et al. Plant foods, dietary fibre and risk of ischaemic heart disease in the European Prospective Investigation into Cancer and Nutrition (EPIC) cohort. Int J Epidemiol 2021; 50(1): 212–22. doi: 10.1093/ije/dyaa155
- 52. Stefler D, Malyutina S, Kubinova R, Pajak A, Peasey A, Pikhart H, et al. Mediterranean diet score and total and cardiovascular mortality in Eastern Europe: the HAPIEE study. Eur J Nutr 2017; 56(1): 421–9. doi: 10.1007/s00394-015-1092-x
- 53. Tong TYN, Appleby PN, Key TJ, Dahm CC, Overvad K, Olsen A, et al. The associations of major foods and fibre with risks of ischaemic and haemorrhagic stroke: a prospective study of 418 329 participants in the EPIC cohort across nine European countries. Eur Heart J 2020; 41(28): 2632–40. doi: 10.1093/ eurheartj/ehaa007
- 54. Yamasaki K, Kayaba K, Ishikawa S. Soy and soy products intake, all-cause mortality, and cause-specific mortality in Japan: The Jichi Medical School Cohort Study. Asia Pac J Public Health 2015; 27(5): 531–41. doi: 10.1177/1010539514539545
- 55. Yu D, Zhang X, Gao YT, Li H, Yang G, Huang J, et al. Fruit and vegetable intake and risk of CHD: results from prospective cohort studies of Chinese adults in Shanghai. Br J Nutr 2014; 111(2): 353–62. doi: 10.1017/S0007114513002328
- Bazzano LA, Li TY, Joshipura KJ, Hu FB. Intake of fruit, vegetables, and fruit juices and risk of diabetes in women. Diabetes Care 2008; 31(7): 1311–7. doi: 10.2337/dc08-0080
- 57. Becerra-Tomás N, Díaz-López A, Rosique-Esteban N, Ros E, Buil-Cosiales P, Corella D, et al. Legume consumption is inversely associated with type 2 diabetes incidence in adults: a prospective assessment from the PREDIMED study. Clin Nutr 2018; 37(3): 906–13. doi: 10.1016/j.clnu.2017.03.015
- Ericson U, Sonestedt E, Gullberg B, Hellstrand S, Hindy G, Wirfält E, et al. High intakes of protein and processed meat associate with increased incidence of type 2 diabetes. Br J Nutr 2013; 109(6): 1143–53. doi: 10.1017/S0007114512003017
- 59. Hodge AM, English DR, O'Dea K, Giles GG. Glycemic index and dietary fiber and the risk of type 2 diabetes. Diabetes Care 2004; 27(11): 2701–6. doi: 10.2337/diacare.27.11.2701
- Khalili-Moghadam S, Mirmiran P, Bahadoran Z, Azizi F. The Mediterranean diet and risk of type 2 diabetes in Iranian population. Eur J Clin Nutr 2019; 73(1): 72–8. doi: 10.1038/ s41430-018-0336-2
- Liu S, Serdula M, Janket SJ, Cook NR, Sesso HD, Willett WC, et al. A prospective study of fruit and vegetable intake and the risk of type 2 diabetes in women. Diabetes Care 2004; 27(12): 2993–6. doi: 10.2337/diacare.27.12.2993
- Meyer KA, Kushi LH, Jacobs DR, Jr., Slavin J, Sellers TA, Folsom AR. Carbohydrates, dietary fiber, and incident type 2 diabetes in older women. Am J Clin Nutr 2000; 71(4): 921–30. doi: 10.1093/ajcn/71.4.921
- O'Connor LE, Hu EA, Steffen LM, Selvin E, Rebholz CM. Adherence to a Mediterranean-style eating pattern and risk of diabetes in a U.S. prospective cohort study. Nutr Diabetes 2020; 10(1): 8. doi: 10.1038/s41387-020-0113-x
- 64. Villegas R, Gao YT, Yang G, Li HL, Elasy TA, Zheng W, et al. Legume and soy food intake and the incidence of type 2 diabetes in the Shanghai Women's Health Study. Am J Clin Nutr 2008; 87(1): 162–7. doi: 10.1093/ajcn/87.1.162
- 65. Yan F, Eshak ES, Shirai K, Dong JY, Muraki I, Tamakoshi A, et al. Soy intake and risk of type 2 diabetes among Japanese

- 66. Abeysekara S, Chilibeck PD, Vatanparast H, Zello GA. A pulsebased diet is effective for reducing total and LDL-cholesterol in older adults. Br J Nutr 2012; 108(Suppl 1): S103–10. doi: 10.1017/S0007114512000748
- Azadbakht L, Kimiagar M, Mehrabi Y, Esmaillzadeh A, Padyab M, Hu FB, et al. Soy inclusion in the diet improves features of the metabolic syndrome: a randomized crossover study in postmenopausal women. Am J Clin Nutr 2007; 85(3): 735–41. doi: 10.1093/ajcn/85.3.735
- Bakhtiari A, Hajian-Tilaki K, Omidvar S, Nasiri-Amiri F. Clinical and metabolic response to soy administration in older women with metabolic syndrome: a randomized controlled trial. Diabetol Metab Syndr 2019; 11: 47. doi: 10.1186/ s13098-019-0441-y
- 69. Cobiac L, McArthur R, Nestel PJ. Can eating baked beans lower plasma cholesterol? Eur J Clin Nutr 1990; 44(11): 819–22.
- Doma KM, Dolinar KF, Dan Ramdath D, Wolever TMS, Duncan AM. Canned beans decrease serum total and LDL cholesterol in adults with elevated LDL cholesterol in a 4-wk multicenter, randomized, crossover study. J Nutr 2021; 151(12): 3701–9. doi: 10.1093/jn/nxab323
- Duane WC. Effects of legume consumption on serum cholesterol, biliary lipids, and sterol metabolism in humans. J Lipid Res 1997; 38(6): 1120–8. doi: 10.1016/S0022-2275(20)37194-7
- Lin BW, Wu ZF, Liu WZ, Zhang RJ. Effects of bean meal on serum cholesterol and triglycerides. Chin Med J (Engl) 1981; 94(7): 455–8.
- 73. Mackay S, Ball MJ. Do beans and oat bran add to the effectiveness of a low-fat diet? Eur J Clin Nutr 1992; 46(9): 641–8.
- 74. Mizelman E, Chilibeck PD, Hanifi A, Kaviani M, Brenna E, Zello GA. A low-glycemic index, high-fiber, ulse-based diet improves lipid profile, but does not affect performance in soccer players. Nutrients 2020; 12(5): 1324. doi: 10.3390/nu12051324
- Nestel P, Cehun M, Chronopoulos A. Effects of long-term consumption and single meals of chickpeas on plasma glucose, insulin, and triacylglycerol concentrations. Am J Clin Nutr 2004; 79(3): 390–5. doi: 10.1093/ajcn/79.3.390
- 76. Pittaway JK, Ahuja KD, Cehun M, Chronopoulos A, Robertson IK, Nestel PJ, et al. Dietary supplementation with chickpeas for at least 5 weeks results in small but significant reductions in serum total and low-density lipoprotein cholesterols in adult women and men. Ann Nutr Metab 2006; 50(6): 512–8. doi: 10.1159/000098143
- 77. Pittaway JK, Ahuja KD, Robertson IK, Ball MJ. Effects of a controlled diet supplemented with chickpeas on serum lipids, glucose tolerance, satiety and bowel function. J Am Coll Nutr 2007; 26(4): 334–40. doi: 10.1080/07315724.2007.10719620
- 78. Saraf-Bank S, Esmaillzadeh A, Faghihimani E, Azadbakht L. Effects of legume-enriched diet on cardiometabolic risk factors among individuals at risk for diabetes: a crossover study. J Am Coll Nutr 2016; 35(1): 31–40. doi: 10.1080/07315724.2014.931262
- Tischmann L, Adam TC, Mensink RP, Joris PJ. Longer-term soy nut consumption improves vascular function and cardiometabolic risk markers in older adults: results of a randomized, controlled cross-over trial. Clin Nutr 2022; 41(5): 1052–8. doi: 10.1016/j.clnu.2022.03.014
- Winham DM, Hutchins AM, Johnston CS. Pinto bean consumption reduces biomarkers for heart disease risk. J Am Coll Nutr 2007; 26(3): 243–9. doi: 10.1080/07315724.2007.10719607
- Winham DM, Hutchins AM. Baked bean consumption reduces serum cholesterol in hypercholesterolemic adults. Nutr Res 2007; 27(7): 380–6. doi: 10.1016/j.nutres.2007.04.017

- Marventano S, Izquierdo Pulido M, Sánchez-González C, Godos J, Speciani A, Galvano F, et al. Legume consumption and CVD risk: a systematic review and meta-analysis. Public Health Nutr 2017; 20(2): 245–54. doi: 10.1017/S1368980016002299
- Martini D, Godos J, Marventano S, Tieri M, Ghelfi F, Titta L, et al. Nut and legume consumption and human health: an umbrella review of observational studies. Int J Food Sci Nutr 2021; 72(7): 871–8. doi: 10.1080/09637486.2021.1880554
- 84. Miller V, Micha R, Choi E, Karageorgou D, Webb P, Mozaffarian D. Evaluation of the quality of evidence of the association of foods and nutrients with cardiovascular disease and diabetes: a systematic review. JAMA Netw Open 2022; 5(2): e2146705. doi: 10.1001/jamanetworkopen.2021.46705
- Li H, Li J, Shen Y, Wang J, Zhou D. Legume consumption and all-cause and cardiovascular disease mortality. Biomed Res Int 2017; 2017: 8450618. doi: 10.1155/2017/8450618
- Deng C, Lu Q, Gong B, Li L, Chang L, Fu L, et al. Stroke and food groups: an overview of systematic reviews and meta-analyses. Public Health Nutr 2018; 21(4): 766–76. doi: 10.1017/ S1368980017003093
- Shi ZQ, Tang JJ, Wu H, Xie CY, He ZZ. Consumption of nuts and legumes and risk of stroke: a meta-analysis of prospective cohort studies. Nutr Met Cardiovasc Dis 2014; 24(12): 1262–71. doi: 10.1016/j.numecd.2014.06.009
- Namazi N, Saneei P, Larijani B, Esmaillzadeh A. Soy product consumption and the risk of all-cause, cardiovascular and cancer mortality: a systematic review and meta-analysis of cohort studies. Food Funct 2018; 9(5): 2576–88. doi: 10.1039/ C7FO01622K
- Nachvak SM, Moradi S, Anjom-Shoae J, Rahmani J, Nasiri M, Maleki V, et al. Soy, soy isoflavones, and protein intake in relation to mortality from all causes, cancers, and cardiovascular diseases: a systematic review and dose-response meta-analysis of prospective cohort studies. J Acad Nutr Diet 2019; 119(9): 1483–500.e17. doi: 10.1016/j.jand.2019.04.011
- 90. Li N, Wu X, Zhuang W, Xia L, Chen Y, Zhao R, et al. Soy and isoflavone consumption and multiple health outcomes: umbrella review of systematic reviews and meta-analyses of observational studies and randomized trials in humans. Mol Nutr Food Res 2020; 64(4): e1900751. doi: 10.1002/mnfr.201900751
- 91. Pearce M, Fanidi A, Bishop TRP, Sharp SJ, Imamura F, Dietrich S, et al. Associations of total legume, pulse, and soy consumption with incident type 2 diabetes: federated meta-analysis of 27 studies from diverse world regions. J Nutr 2021; 151(5): 1231–40. doi: 10.1093/jn/nxaa447

- 92. Tang J, Wan Y, Zhao M, Zhong H, Zheng JS, Feng F. Legume and soy intake and risk of type 2 diabetes: a systematic review and meta-analysis of prospective cohort studies. Am J Clin Nutr 2020; 111(3): 677–88. doi: 10.1093/ajcn/nqz338
- Bazzano LA, Thompson AM, Tees MT, Nguyen CH, Winham DM. Non-soy legume consumption lowers cholesterol levels: a meta-analysis of randomized controlled trials. Nutr MetabCardiovasc Dis. 2011; 21(2): 94–103. doi: 10.1016/j. numecd.2009.08.012
- 94. Ha V, Sievenpiper JL, de Souza RJ, Jayalath VH, Mirrahimi A, Agarwal A, et al. Effect of dietary pulse intake on established therapeutic lipid targets for cardiovascular risk reduction: a systematic review and meta-analysis of randomized controlled trials. CMAJ. 2014; 186(8): E252–62. doi: 10.1503/ cmaj.131727
- 95. Ferreira H, Vasconcelos M, Gil AM, Pinto E. Benefits of pulse consumption on metabolism and health: a systematic review of randomized controlled trials. Crit Rev Food Sci Nutr 2021; 61(1): 85–96. doi: 10.1080/10408398.2020.1716680
- 96. Bielefeld D, Grafenauer S, Rangan A. The effects of legume consumption on markers of glycaemic control in individuals with and without diabetes mellitus: a systematic literature review of randomised controlled trials. Nutrients 2020; 12(7): 2023. doi: 10.3390/nu12072123
- Heidemann C, Hoffmann K, Spranger J, Klipstein-Grobusch K, Möhlig M, Pfeiffer AF, et al. A dietary pattern protective against type 2 diabetes in the European Prospective Investigation into Cancer and Nutrition (EPIC) Potsdam Study cohort. Diabetologia 2005; 48(6): 1126–34. doi: 10.1007/s00125-005-1743-1
- Rees K, Takeda A, Martin N, Ellis L, Wijesekara D, Vepa A, et al. Mediterranean-style diet for the primary and secondary prevention of cardiovascular disease. Cochrane Database Syst Rev. 2019; 3(3): CD009825. doi: 10.1002/14651858.CD009825.pub3
- Verbeek J, Hoving J, Boschman J, Chong LY, Livingstone-Banks J, Bero L. Systematic reviews should consider effects from both the population and the individual perspective. Am J Public Health 2021; 111(5): 820–5. doi: 10.2105/AJPH.2020.306147

*Birna Thorisdottir

Faculty of Food Science and Nutrition University of Iceland Aragata 14, 102 Reykjavik, Iceland Email: bith@hi.is